Author’s response to reviews

Title: AN ATYPICAL AND BILATERAL PRESENTATION OF CHARCOT FOOT DISEASE

Authors:
Chariclia Loupa (chariclia.loupa@gmail.com)
E. Meimeti (e.meimeti@gmail.com)
A. Kokas (gram_dioikhshs@flemig-hospital.gr)
E.D. Voyatzoglou (podiatry21@gmail.com)
A. Donou (andriana.donou@yahoo.gr)

Version: 3 Date: 10 Aug 2019

Author’s response to reviews:

We note that you have included author EK in your authors' contributions section but this author is not identifiable in the submission system. We also note that author A. Kokas is missing from the authors' contribution section. Please ensure the individual contributions of all authors are included in the authors' contributions section.

- DONE

2. Please include an abstract after your title page. Your abstract should include the following headings: Background, Case Presentation, Conclusions, and should be no longer than 350 words. Please also include a keywords section, which should three to ten keywords.

- DONE

3. In order to be in line with journal requirements, please include the following headings within your manuscript: Background, Case Presentation, Discussion and Conclusion (where Discussion and Conclusion are one section).

- DONE
4. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

- DONE